Author's response to reviews

Title: A retrospective cohort study on the influence of UV index and race/ethnicity on risk of stress and lower limb fractures

Authors:

Scott J Montain (scott.montain@us.army.mil)
Susan M McGraw (susan.mcgraw@us.army.mil)
Matthew R Ely (matthew.ely@us.army.mil)
Tyson L Grier (tyson.grier@us.army.mil)
Joseph J Knapik (joseph.knapik@us.army.mil)

Version: 4 Date: 3 April 2013

Author's response to reviews:

Per editor request of 31 March 13, we have 1) updated the methods section so that it now provides the name of the ethics committee that approved the study, and 2) added a Competing Interests subsection and disclosure statement. These are underlined and in red text.

Scott Montain